Author's response to reviews

Title: Enzyme-linked immunoassay for dengue virus IgM and IgG antibodies in serum and filter paper blood

Authors:

Nga T.T. Tran (ngatran_crh@yahoo.com)
Peter J. de Vries (p.j.devries@amc.uva.nl)
Lan P. Hoang (lanphuongh@hcm.vnn.vn)
Giao T. Phan (giaothaoh@hcmc.netnam.vn)
Hung Q. Le (equochung@saigonnet.vn)
Binh Q. Tran (binhtq@hcm.vnn.vn)
Mai T.C. Vo (chimai@hcm.vnn.vn)
Nam V. Nguyen (sotretbt@yahoo.com)
Jan Groen (jgroen@focustechnologies.com)
Nico Nagelkerke (niconagelkerke@hotmail.com)
Piet A. Kager (p.a.kager@amc.uva.nl)

Version: 3 Date: 13 August 2005

Author's response to reviews:

No changes were made since the manuscript was transferred by BiomedCentral from microbiology to infectious diseases. This is the original manuscript and not a revised version.